

Dear Dr. Editorial Board,

We are very honored to have the opportunity to publish a paper in your journal. We revised our paper based on peer review and the science editor. The detailed modifications are as follows.

To reviewer #1:

Present review also explained and compared kinds of specific endoscopic resections, and explored which method are suggested for sGIST. But the abstract and core tips part have not manifest that. The “<”mark in page 6 is not the English symbol.

Response: Thank you very much for this valuable feedback. We have added the selection of endoscopic treatment methods in the abstract and core tips part of the revised draft. And all the symbols have been changed to English symbol.

To the science editor:

(1) The authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s); (2) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor; and (3) If an author of a submission is re-using a figure or figures published elsewhere, or that is copyrighted, the author must provide documentation that the previous publisher or copyright holder has given permission for the figure to be re-published; and correctly indicating the reference source and copyrights.

Response: We sincerely appreciate this valuable feedback. We will upload the contract documents of the funding agency and all provide original pictures in a PowerPoint. All pictures are not reused and are not published elsewhere.

In addition, I'm really sorry that due to our negligence, the first two authors' names (**Ze-Ming Chen, Min-Si Peng**) were entered with an extra "h" consonant, and now we have corrected them together.

We sincerely hope that this revision will meet the expectations of the science editor and reviewer. Please feel free to contact us if further modifications are required.

Thank you very much!

Sincerely,

Zheng-Lei Xu, MD